Author's response to reviews

Title: Internet-based Self-Assessment after the Tsunami: lessons learned

Authors:

Stefan Vetter (stefan.vetter@access.uzh.ch)
Astrid Rossegger (astrid.rossegger@me.com)
Thomas Elbert (thomas.elbert@uni-konstanz.de)
Juliane Gerth (juliane.gerth@gmail.com)
Frank Urbaniok (frank.urbaniok@ji.zh.ch)
Arja Laubacher (arja.laubacher@ji.zh.ch)
Wulf Rossler (roessler@dgsp.uzh.ch)
Jérôme Endrass (jerome.endrass@me.com)

Version: 8 Date: 20 December 2010

Author's response to reviews:

We have completed all of the major and minor revision required by the editorial production team.